Author's response to reviews

Title: Prevalence and determinants of intimate partner violence towards female students of the University of Ibadan, Nigeria.

Authors:

Joseph E Umana (umanajoe@yahoo.com)
Olufunmilayo I Fawole (fawoleo@ymail.com)
Ikeola A Adeoye (drnichike@yahoo.com)

Version: 3
Date: 29 January 2014

Author's response to reviews:

The Executive Editor

BMC series journal

Biomed Central

28/1/14

Response to the concerns of our manuscript’s reviewer - Prevalence and correlates of intimate partner violence towards female students of the University of Ibadan, Nigeria

Response to Katie Edwards:

1. The authors have replaced wife beating and interpersonal violence with IPV. See page 4, paragraph 2, line 1; page 3, paragraph 4, line 1; and page 10, paragraph 1, lines 3 and 4.

2. Cigarette smoking, alcohol use, etc have been tempered as correlates. See page 4, paragraph 2, line 1; page 4, paragraph 3, line 1; and page 4, paragraph 4, line 2.

3. More description for the method on measures has been included. Please see page 5, last paragraph.

4. The entire discussion section has been rewritten. It is now more organized and flows.
5. The typographical errors and writing style issues have been addressed.

Thank you

Joe Umana